Author’s response to reviews

Title: Immune-enhancing activity of C. militaris fermented with Pediococcus pentosaceus (GRCON89A) in CY-induced immunosuppressed model

Authors:

Ha-Kyoung Kwon (kyoungkh93@naver.com)
Woo-Ri Jo (jjori515@naver.com)
park hye-jin (nimpi79@hanmail.net)

Version: 1 Date: 15 Dec 2017

Author’s response to reviews:

Dear. Editors

Dear Dr. Messin

We greatly appreciate the efforts of the Editorial Board and reviewers for giving us the valuable comments on our manuscript for publication in BMC Complementary and alternative medicine. Here, we have enclosed a revised version of manuscript in response to the reviewer comments. Included below is a point-by-point description of our responses to the Editor’s and reviewer’s comments. The revision made in the revised note was written in blue. We cordially hope that you and the reviewers find this revised manuscript acceptable for publication in BMC Complementary and alternative medicine.

Major changes

1. The numbering and organization of figures were changed as summarized in below.

Revised figure and table number.

Figure 1 > Figure 3A and Figure 4A
Figure 2 > Figure 1
2. In the revised manuscript, the immune regulatory activity of fermented C.militaris grown on germinated rhynchosia nulubilis (GRC-ON89A) on macrophage activity was statistically compared to that of unfermented C.militaris grown on germinated rhynchosia nulubilis (GRC)

Sincerely,

Hye-Jin Park